Author’s response to reviews

Title: Risk factors of one year increment of heart calcifications and survival in hemodialysis patients

Authors:

Giorgio Coen (giorgio.coen@gmail.com)
Andrea Pierantozzi (andrea.pierantozzi@virgilio.it)
Daniele Spizzichino (daniele.spizzichino@uniroma1.it)
Daniela Sardella (daniela.sardella@fastwebnet.it)
Daniela Mantella (danielamantella@gmail.com)
Micaela Manni (micaela.manni@tin.it)
Andrea Romagnoli (romagnoli69@libero.it)
Roberta Pacifici (roberta.pacifici@iss.it)
Piergiorgio Zuccaro (zuccaro@iss.it)
Salvatore Di Giulio (digiulio@tin.it)
Luigi Pellegrino (luigi.pellegrino75@libero.it)

Version: 5 Date: 16 April 2010

Author’s response to reviews: see over
Dear Sir,

enclosed is the revised manuscript “Risk factors of one year increment of coronary calcifications and survival in hemodialysis patients”.

The manuscript has been revised for english by an english teacher. In addition, the manuscript is now conform to the journal style.

The sentence:

*Ethical approval of the study was not required for its observational nature.*

has been reported in the Methods

Changes to the manuscript are shown in italics.

Best regards,

Giorgio Coen, MD